

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Blattner, Laura; Farinha, Luísa A.; Nogueira, Gil

Working Paper Not all shocks are created equal: Assessing heterogeneity in the bank lending channel

ECB Working Paper, No. 2607

Provided in Cooperation with: European Central Bank (ECB)

Suggested Citation: Blattner, Laura; Farinha, Luísa A.; Nogueira, Gil (2021) : Not all shocks are created equal: Assessing heterogeneity in the bank lending channel, ECB Working Paper, No. 2607, ISBN 978-92-899-4860-9, European Central Bank (ECB), Frankfurt a. M., https://doi.org/10.2866/200470

This Version is available at: https://hdl.handle.net/10419/246184

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Working Paper Series

Laura Blattner, Luísa Farinha, Gil Nogueira

Not all shocks are created equal: assessing heterogeneity in the bank lending channel

ECB – Lamfalussy Fellowship Programme



Disclaimer: This paper should not be reported as representing the views of the European Central Bank (ECB). The views expressed are those of the authors and do not necessarily reflect those of the ECB.

ECB Lamfalussy Fellowship Programme

This paper has been produced under the ECB Lamfalussy Fellowship programme. This programme was launched in 2003 in the context of the ECB-CFS Research Network on "Capital Markets and Financial Integration in Europe". It aims at stimulating high-quality research on the structure, integration and performance of the European financial system.

The Fellowship programme is named after Baron Alexandre Lamfalussy, the first President of the European Monetary Institute. Mr Lamfalussy is one of the leading central bankers of his time and one of the main supporters of a single capital market within the European Union.

Each year the programme sponsors five young scholars conducting a research project in the priority areas of the Network. The Lamfalussy Fellows and their projects are chosen by a selection committee composed of Eurosystem experts and academic scholars. Further information about the Network can be found at http://www.eufinancial-system.org and about the Fellowship programme under the menu point "fellowships".

Abstract

We provide evidence that the strength of the bank lending channel varies considerably across three major events in the European sovereign debt crisis – the Greek debt restructuring (PSI), outright monetary transactions (OMT), and quantitative easing (QE). We study how lending responds to each shock using detailed bank, firm, and household data from Portugal, a country that was directly exposed to the three events. While the price of sovereign debt securities increased in all three events, banks reduced sovereign debt holdings and realized accumulated capital gains only after QE. As a result, lending to final borrowers reacted more strongly to QE than to the PSI or OMT events. Our results suggest that asset purchases were more effective than signalling events at stimulating the bank lending channel.

Keywords: asset purchases, bank lending channel, OMT, PSI, QE **JEL codes:** E52, E58, G18, G21

Non-technical Summary

The bank lending channel theorizes that monetary policy events affect lending to firms and households through banks. Central banks may use this channel to incentivize lending in periods of credit market distress. However, evidence of the effectiveness of the bank lending channel is still scarce. Some papers have studied the effects of individual events in isolation, but it is difficult to compare different policy interventions because of the use of different samples and techniques. We add to this literature (e.g., Di Maggio et al. (2016), Chakraborty et al. (2017)) by studying events that affected bank balance sheets in the European sovereign debt crisis jointly and documenting significant heterogeneity in the strength of the bank lending channel.

We compare the effects of the announcement of three major events in the European sovereign debt crisis: the Greek Private Sector Involvement (PSI) in February 2012, Outright Monetary Transactions (OMT) in July 2012, and quantitative easing (QE) in January 2015. These interventions sought to revive the euro area economy, especially in countries exposed to the European sovereign debt crisis. As we and other researchers show using security price data (e.g., Koijen et al. (2017a), Krishnamurthy et al. (2018)), the three events had a positive effect on the price of European government bonds. However, QE differed from PSI and OMT because the central bank only conducted asset purchases with QE. We analyze the effects of these events using credit register, bank, and firm data covering 44 Portuguese banks and their credit relationships with firms and households.

We start our analysis by estimating the effect of exposure to euro area sovereign debt securities on bank balance sheets. We find that banks with a larger exposure to government bonds reduced their sovereign debt holdings in the first year after QE, but not after PSI and OMT. Next, we analyze the transmission of the three events to firms and households. In the first year after QE, banks that were more exposed to government bonds lent more to their customers, were less likely to terminate credit relationships, and were more likely to accept loan applications. We also find evidence of changes in creditor composition for QE, as the effect of exposure to sovereign debt on lending was stronger for riskier firms. We find a more moderate to null effect for PSI and OMT. Finally, we use detailed financial statement data to analyze the effect of exposure to sovereign debt on aggregate firm outcomes. Firm exposure to sovereign debt securities had a largely positive effect on aggregate firm-level lending after QE and a milder effect after PSI and OMT. Additionally, exposure to sovereign debt had a positive effect on sales growth for firms that applied for new loans at exposed banks after QE. Overall, we provide evidence that QE had a stronger effect than PSI and OMT banks increased lending more after QE, and to some extent borrowers responded to additional credit availability with more economic activity. Overall, our evidence suggests that asset purchases were more effective than signaling events at stimulating the bank lending channel during the European sovereign debt crisis.

1 Introduction

In macro-finance models with financial intermediation, banks affect the transmission of monetary policy to final borrowers through the bank balance sheet channel. Banks hold securities in their balance sheet and increase lending when the value of these securities goes up. Interventions by the policymaker affect security prices and cause variation in bank lending. Shocks that potentially affect lending are not all equal. In some cases, the policymaker just signals its future behavior. In other cases, it intervenes directly by purchasing assets. Empirically, there is little evidence on how elasticities vary with the nature of shocks. Some papers study the effect of specific events on lending in isolation. However, it is difficult to compare different policies because the samples and techniques researchers use vary across papers.

In this paper, we fill this gap by comparing the effect of three major events of the European sovereign debt crisis on banks and borrowers: the Greek debt restructuring (PSI), Outright Monetary Transactions (OMT), and Quantitative Easing (QE). We use a sample of 44 Portuguese banks that were directly exposed to these shocks. Portugal provides a unique laboratory to test the effect of different shocks on the bank lending channel as the country was exposed directly to the three events. PSI avoided a disorderly Greek default (Zettelmeyer et al. (2013)). Positive news from Greece signaled lower risk of contagion and reduced government bond yields in other European periphery countries such as Portugal (Mink and De Haan (2013)). OMT directly affected Portuguese government bonds. Ireland and Portugal were the only countries that were eligible for OMT purchases because they received a financial assistance program but did not default on their debt. The yields of Portuguese bonds dropped significantly because of the OMT announcement (Krishnamurthy et al. (2018)). QE also had a positive effect on the price of sovereign debt securities from Portugal and other European countries (Koijen et al. (2017a)). However, in contrast to PSI and OMT, the central bank purchased government bonds under the QE program.

Using detailed bank, household, and firm data, we measure the effect of exposure to sovereign debt securities on bank balance sheets and lending for the three events. First, we use ISIN-level data with detailed information on the price, quantity, and book value of all securities held by individual banks to analyze the effect of each shock on bank balance sheets. This data allows us to compute the exposure of each bank to European sovereign debt. The value of securities held by exposed banks increased more after PSI and OMT than after QE. However, exposed banks reduced their exposure to government bonds and realized accumulated capital gains after QE but not after PSI and OMT.

Second, we analyze the transmission of these shocks to households and firms in the intensive and extensive margins. Using credit register data, we compute the change in lending for each borrower-bank credit relationship between the month before the event and up to one year after the event. Using the identification strategy from Khwaja and Mian (2008), we control for the effect of borrower demand on lending. We also compute the success rate of loan applications by employing a methodology that is similar to the one used by Jiménez et al. (2012). After QE, exposed banks lent more to their customers, were less likely to terminate credit relationships, and were more likely to accept loan applications. We also find evidence of changes in creditor composition for QE, as the effect of exposure to sovereign debt on lending was stronger for riskier firms. We find that exposure to sovereign debt had a more moderate to null effect on lending with PSI and OMT. We measure the elasticity of bank lending to changes in security prices and capital gains. We do not observe a strong relationship between price appreciation and lending. However, we find evidence of a strong and positive relationship between lending and realized capital gains, particularly after QE.

Finally, we analyze the effect of exposure to sovereign debt on aggregate firm outcomes. Consistent with the results from the bank-borrower analysis, exposure to sovereign debt securities had a largely positive effect on aggregate firm-level lending after QE and a milder effect after PSI and OMT. Additionally, exposure to sovereign debt had a positive effect on sales growth for firms that applied new loans at exposed banks after QE. We find no evidence that exposure to government debt had a positive effect on real outcomes for firms with existing relationships in any of the three events or for new loan applicants after PSI and OMT.

The contribution of this paper is twofold. First, we show that in a unified framework that covers the universe of borrowers in an economy, balance sheet shocks have a different effect on the transmission of lending to households. Policies such as PSI and OMT had a limited effect on lending in the intensive and extensive margins through the balance sheet channel, while asset purchases had a strong positive impact. Second, we compute lending elasticities to asset prices and capital gains. With these findings, we inform quantitative macro-finance modeling and show that bank shocks do not affect the bank lending channel in the same way.

Literature Our research contributes to the literature on the bank lending channel of monetary policy (Kashyap and Stein (1994), Holmstrom and Tirole (1997)), specifically to a growing body of literature exploiting variation in banks' sovereign debt holdings. We analyze empirically the effect of policies that potentially affect lending through changes in the value of assets held by banks (Brunnermeier and Sannikov (2016)), or through incentives to rebalance bank portfolios with assets that have similar characteristics (Goldstein et al. (2018)). Existing papers study single episodes in the sovereign debt markets in isolation¹. For example, Bottero et al. (2019) study the corporate credit supply effects of the onset of the sovereign debt crisis in Italy. Acharya et al. (2017) exploit heterogeneity in the banks' sovereign bond holdings to study the effect of the announcement of the ECB's Outright Monetary Transactions (OMT) on credit misallocation to zombie firms. Our contribution is to study events that affect bank balance sheets jointly and document significant heterogeneity in the strength of the bank lending channel. We also relate our findings to the trading behavior of banks.

We also contribute to the extensive literature on the transmission and effects of unconventional monetary policy. In the US, Di Maggio et al. (2016), Chakraborty et al. (2017), Luck and Zimmermann (2017) and Rodnyansky and Darmouni (2017) exploit the differential bank holdings of mortgage-backed securities—similar to our cross-sectional identification strategy—find positive the effects of QE on lending in the mortgage market, especially for QE3, the most similar to the ECB's QE program.² Similar to our findings, Luck and Zimmermann (2017) find positive effects on both

 $^{^{1}}$ The literature also studies other ECB interventions, predominantly long-term refinancing operations (LTRO). Carpinelli and Crosignani (2017) study the ECB liquidity injections (LTRO) 2011-2012.

 $^{^{2}}$ The Federal Reserve's QE3 and the ECB EAPP were both open-ended commitments to purchase a certain monthly volume of securities until economic conditions improved. In earlier work, Stroebel and Taylor (2012) analyze the effect of the Federal Reserve's mortgage-backed securities purchase program on mortgage spreads.

commercial and industrial (C&I) and household lending.³ Beyond the US, Morais et al. (2019) study the international transmission of quantitative easing through foreign banks. In Europe, the literature on the real effects of QE has focused on the real effects of corporate-sector purchase programs (e.g. B Grosse-Rueschkamp, S Steffen, D Streitz (2017), see also Dell'Ariccia et al. (2018) for a survey).

Finally, our paper is related to literature that studies the incentives of European banks when choosing exposure to sovereign debt. Acharya and Steffen (2015) argue that a combination of regulatory arbitrage and risk-shifting led banks to increase exposure to sovereign debt with the onset of the sovereign debt crisis. Becker and Ivashina (2017) and Ongena et al. (2019) argue that government pressure played a role in the banks' increasing exposure to domestic sovereign debt. Abbassi et al. (2016) show that banks with trading desks scaled up their investment into sovereign debt as the yields became more attractive with the onset of the sovereign debt crisis. Our contribution is to show that these forces matter for the transmission of unconventional policy to lending. We show that banks only react to valuation gains by reducing their net positions in response to QE when many of the above incentives had been mitigated. We confirm the finding of Koijen et al. (2017b) that Euro-area banks started selling eligible securities to the ECB in response to QE.

The remainder of this paper is organized as follows. Section 2 describes the data and the sovereign debt events we exploit. Section 3 describes the empirical strategy. Section 4 provides results. Section 5 concludes.

2 Background and Data

In this section, we provide background on the three episodes we study and describe our data.

 $^{^{3}}$ In earlier work, Chakraborty et al. (2017) found that the increase in credit supply in the US mortgage market crowds out C&I lending.

2.1 Background on Events in European Sovereign Debt Markets

We study three key events that induced large changes in sovereign debt prices: the Greek government PSI program in February 2012, the OMT announcement in July 2012, and finally the announcement of the ECB's QE program in January 2015.

PSI On April 23 2010, the Greek government officially requested a bailout package from the EU and IMF, triggering a series of events that culminated in the European sovereign debt crisis (Lane, 2012). In 2012, there were two major interventions in the European securities market as a direct response to the sovereign debt crisis. In late February, the Greek government announced the *private sector involvement* (PSI), a program to restructure sovereign debt held by private investors. Under this agreement, bond holders accepted a haircut of 53.5% in exchange for new Greek government debt securities. On 9 March 2012, the International Swaps and Derivatives Association (ISDA) declared that Greece's restructuring represented a default, implying that credit default swaps would be triggered. The Greek government introduced a retroactive collective action clause to enforce private sector participation. PSI is widely seen as a positive system-wide shock as the restructuring prevented a potential disorderly default (e.g., Zettelmeyer et al. (2013)). Portuguese yields peaked in the month before PSI was announced and declined steadily in the subsequent months (Figure 1). In Appendix A we estimate that PSI reduced Portuguese government bond yields by 160-420 basis points.

[Figure 1]

OMT In July 2012, the then-president of the ECB, Mario Draghi, gave a speech in London that included the famous remark that "within our mandate, the ECB is ready to do whatever it takes to preserve the euro. And believe me, it will be enough." This speech was succeeded a week later by the *Outright Monetary Transactions* (OMT) program, in which the ECB pledged to buy government bonds from bailed out countries. This program was subject to conditionality: governments of bailed out countries could not default on their debt and had to actively request intervention from the ECB.

Even though the ECB never made use the OMT program, the OMT was perceived as having been successful in bringing down bond yields (Krishnamurthy et al. (2018)).

QE In January 2015, the ECB announced the *Public Sector Purchase Program* (PSPP), the first major government bond quantitative easing program in Europe. The ECB pledged to buy government bonds in the secondary market every month. The objective of these purchases was to decrease yields at the long end and transmit monetary policy to final borrowers through greater lending. PSPP distinguished itself from OMT for two reasons. First, it increased prices of government bonds not only from the periphery but also from countries in the core of the Euro Area (Koijen et al. (2017a)). Second, it resulted in actual purchases of assets from the central bank.

2.2 Portugal as a Laboratory

Portugal provides an excellent laboratory to study key events in sovereign debt markets that affected European banks in the aftermath of the sovereign debt crisis.

At the onset of the crisis, the balance sheets of Portuguese banks were directly exposed to securities from periphery countries (Greece, Ireland, Italy, Portugal, and Spain). Throughout the crisis period, banks increased their exposure to Portuguese securities significantly. Figure 2 depicts the evolution of sovereign debt held by Portuguse banks. In January 2010, sovereign debt from Portugal represented about 2% of the assets held by banks in the sample. At the end of 2016, Portuguese debt securities represented almost 6% of the assets held by these banks. The exposure to sovereign debt securities from other periphery countries was about 1% in the beginning of 2011 and increased to approximately 3% at the end of 2016.

[Figure 2]

Portugal was also directly exposed to each of the major programs that affected bond prices. Portugal and Ireland were the only countries that were eligible for OMT purchases, which required participants to be under an adjustment program and to have no default history. Among periphery countries, Portugal was also the most exposed to sovereign debt events in Greece (Mink and De Haan (2013)). Finally, the ECB actively bought Portuguese government bonds from Portuguese banks through the implementation of the QE program (Koijen et al. (2017a)).

2.3 Data

We use data for the universe of banks that operate continuously in Portugal between April 2010 and January 2016. Table 1 shows descriptive statistics for the variables used in the analysis for each of the 44 banks across the three events. In Columns (1) and (2) of Table 1a, we show descriptive statistics weighted by the number of borrowers in the sample (including borrowers with single relationships). In Columns (3) and (4), we show unweighted statistics. In Table 1b we provide descriptive statistics for firm and household borrowers. We show statistics for households with multiple credit relationships, as we use borrower fixed effects to control for demand factors throughout our analysis (see Khwaja and Mian (2008)). In Table A1 we provide descriptive statistics separately for each event.

 $[\mbox{ Table 1 }]$

2.3.1 Credit Register

We obtain monthly lending data for all banks in Portugal from the Portuguese credit register maintained by the Bank of Portugal. The credit register has two distinctive advantages. First, there is nearly universal coverage because banks must report all exposures above EUR 50. Second, it covers both households and non-financial firms. Household loans include credit card borrowing, personal loans, and mortgage loans. An important feature of the data is that a substantial fraction of nonfinancial firms and households have multiple lending relationships. This allows us to implement a borrower fixed effect strategy following Khwaja and Mian (2008).

2.3.2 Credit Registry Inquiries

We also draw on the database of bank inquiries to the credit registry maintained by the Bank of Portugal. Banks can request information on a potential borrower from the credit register. Inquiries are usually requested only for a loan application of a potential new corporate or household borrower by lenders that have no pre-existing credit relationship. We match the inquiry data to the credit registry to compute successful and unsuccessful loan applications following Jiménez et al. (2012).

2.3.3 Bank Data

We use four separate sources of bank data from the Bank of Portugal. At the ISIN level, we use SIET (Estatísticas de Emissões de Títulos), a dataset that contains the list of all securities (with ISIN) held by each bank at the monthly frequency. We also use an *Asset Purchase Program* (APP) purchases dataset. This file contains the list of transactions of Portuguese assets performed by the ECB under the APP and includes the identification of counterparties. This dataset is provided by the Bank of Portugal.

We retrieve bank balance sheet variables from MFI (Estatísticas Monetárias e Financeiras), which has monthly financial statements for all banks. We augment this dataset with quarterly regulatory ratios from an additional FINREP/COREP dataset

Total gains with securities are derived by multiplying monthly price variation by outstanding security holdings. Realized gains are cumulative gains that banks monetize when they close their position in a certain security. We compute the effect of price variation and capital gains on bank net worth using the following formula:

$$\Delta \log net worth_{bk} = \log(1 + capital_{bt} + \Delta value_{bk}) - \log(1 + capital_{bt}) \tag{1}$$

Where $capital_{bt}$ is the capital and reserves of bank b in the last period before the event and $\Delta value_{bk}$ is the change in value or capital gains for securities in SIET. We set $\Delta \log net \ worth = -1$ when $-\Delta value_{bk} > capital_{bt}$. We winsorize monthly price changes from SIET at the 1% level.

We define bank exposure to sovereign securities using ISIN-level data from SIET. We obtain a list of marketable securities that are eligible for collateral from the ECB website at the monthly frequency. Securities from SIET are classified as exposed sovereign debt if they observe three conditions: 1) appear at least once in the list of marketable securities from the ECB; 2) are issued by central governments and; 3) are denominated in euro.

2.3.4 Firm Data

We use annual firm financial statements from Base de Dados das Contas Anuais (BDCA), a repository of financial statements maintained by the Portuguese government that covers the universe of non-financial corporations. We retrieve CAPEX/Sales and interest rates from this dataset and winsorize these variables at the 1% level.

3 Empirical Specification

We estimate the following specification for the cross-section of banks at each event:

$$y_b = \beta_1 exposure_b + \gamma X_b + \epsilon_b \tag{2}$$

where y_i is the dependent variable of interest. We obtain $exposure_b$ by standardizing $\frac{Exposed assets_b}{Total bank assets_b}$ the book value of exposed assets divided by total assets in the last period before the event. We also include a number of bank controls X_i to absorb any confounding changes from changes in size (total assets), capitalization (capital ratio), or different levels of non-performing loans. The validity of our approach relies on the the assumption that the degree of a bank's exposure to sovereign debt should not be correlated with any other contemporaneous shock. One potential confounding factor are the ECB's long-term refinancing operations (LTRO) that provided cheap financing to the European banking system (e.g., Carpinelli and Crosignani (2017)). To address this concern, we include the bank-level LTRO take-up in X_b .

We also estimate a bank-level dynamic differences-in-differences model:

$$y_{bt} = \alpha_b + \alpha_t + \sum_{k=2011Q4}^{2016Q4} \beta_k(exposure_b \times \mathbb{1}_{t=k}) + \epsilon_{bt}$$
(3)

where y_{bt} is the outcome of interest in period t, α_b and α_t are bank and time fixed effects, and $\mathbb{1}_{t=k}$ is an indicator variable that is equal to 1 in period k.

At the borrower(i)-bank(b) level, we estimate the following specification cross-sectionally for each event:

$$y_{bi} = \beta_1 \text{exposure}_b + \theta_i + \varphi X_{bi} + \epsilon_{bi} \tag{4}$$

where y_{bi} is the outcome of interest. θ_i is a borrower fixed effect (Khwaja and Mian (2008)) that absorbs any borrower-level changes in credit growth, allowing us to identify the effect of changes in the share of lending supplied by more versus less exposed banks. We include borrower-bank level controls X_{bi} , including the number of years the relationship has existed to address worries that longer relationship might receive a differential treatment from recently established lending relationships. We cluster standard errors at the bank level, which is our level of the treatment (Bertrand et al. (2004)).

In addition to the lending volume in existing lending relationships (intensive margin), we also estimate effect of sovereign debt exposure on the likelihood that an existing relationships ends (linear probability model) and on new loan approval rates using credit registry inquiries (as in Jiménez et al. (2012)). Successful applications are credit inquiries followed by a lending relationship in the subsequent 6 months.

Additionally, we provide direct estimates of the lending elasticity to changes in bank net worth induced by price increases or realized capital gains:

$$y_{bi} = \tilde{\beta}_1 \Delta \text{net worth}_b^{capital} + \tilde{\beta}_2 \Delta \text{net worth}_b^{price} + \theta_i + \gamma X_{bi} + \nu_{bi}$$
(5)

where $\Delta \text{net worth}_{b}^{price}$ and $\Delta \text{net worth}_{b}^{capital}$ are changes to bank net worth induced by price appreciation and capital gains (see Equation 1).

In the Appendix (Table A11) we also estimate an instrumental variables model using 2SLS. We estimate the following first stage equation:

$$\Delta \text{net worth}_{b}^{capital} = \pi exposure_{b} + \theta_{i} + \delta \gamma X_{bi} + \nu_{bi}$$
(6)

In the second stage, we estimate:

$$y_{bi} = \bar{\beta}_1 \Delta \text{net worth}_b^{capital} + \theta_i + \gamma X_{bi} + \epsilon_{bi} \tag{7}$$

where we instrument for the bank-level change in net worth with the bank-level exposure using estimates from Equation 6. The 2SLS coefficient provides the elasticity estimates typically required to calibrate quantitative models.

Finally, we also estimate the effect of exposure to government debt securities on firm characteristics. In Equation 4 we can control for demand factors using borrower fixed effects θ_i . We cannot do the same after aggregating outcomes at the firm level. Instead, we control for time-varying factors that may affect firm demand by interacting firm fixed effects with a linear time trend. We estimate the following equation:

$$y_{it} = \alpha_i + \alpha_t + \sum_m \lambda_m [(t+6) \times \mathbb{1}_{i=m}] + \beta(exposure_b \times \mathbb{1}_{t\geq 0}) + \epsilon_{it}$$
(8)

Where y_{it} is a firm-level outcome t years after the event, α_i and α_t are firm and year fixed effects, $\mathbb{1}_{i=m}$ is an indicator variable equal to 1 for firm i and 0 otherwise, and $\mathbb{1}_{t\geq 0}$ is an indicator variable equal to 1 for observations in the year of the event or subsequent years.

4 Results

We describe the results of our paper in this section. First, we discuss the effect of exposure to sovereign debt securities on bank balance sheets for PSI, OMT, and QE. Second, we discuss the transmission of bank balance sheet shocks to firms and households.

4.1 Banks

In Table 2, we measure the effect of exposure to European sovereign debt on changes in exposed assets after the announcement of PSI, OMT and QE. The dependent variable $\frac{\Delta Exposed assets_k}{Total Assets_t}$ measures the change in exposed assets after each event as a percentage of bank assets in the month before the event. We compute changes in holdings by multiplying changes in quantity by the price in the month before the event. We do this to filter out price effects from changes in holdings. In Table 2, we report results for a one-year horizon after each event (includes the month of the event and the subsequent twelve months). In Table A2, we repeat the analysis at shorter horizons – month and quarter. The month horizon includes the month of the announcement and the subsequent three months. In Table 2 and throughout the rest of the paper, we use the number of customers as sample weights in bank-level regressions. We provide coefficient estimates using unit weights in Table A2.

[Table 2]

More exposed banks did not reduce their exposure to government bonds after OMT and PSI. After QE, banks that were more exposed to government bonds reduced their exposure relatively more. A one standard deviation increase in the exposure to government bonds is associated to a subsequent 1.9 p.p. decrease in government bond holdings. Before the event, the average exposure to government bonds was 5.1% (Table A1), meaning that the coefficient is relatively large.

In Figure 3, we depict estimates for β_k from Equation 3 between the first quarter of 2014 and the last quarter of 2015. W use the cumulative variation in debt holdings as the dependent variable. Consistent with QE having an effect on sovereign debt holdings, exposed banks started reducing their exposure more in the quarter when QE was announced.

[Figure 3]

Banks registered in Portugal were active in selling securities under the Quantitative Easing Program. Figure 4 depicts the percentage of Portuguese securities in the Quantitative Easing that were sold by banks in the sample. These institutions often represented over 20% of the sales.

[Figure 4]

In Table 3, we establish a relationship between exposure to sovereign debt and changes in bank net worth attributable to price appreciation and capital gains. One additional standard deviation in exposure increases price appreciation by 18% with PSI, 6% with OMT, and 3% with QE. However, there is a negligible relationship between capital gains and exposure with PSI and OMT, while one standard deviation in exposure increases capital gains by 2% with QE. These results are consistent with the results from Table 3, as exposed banks only reduce their security holdings after QE. In Tables A3 and A4 we repeat the analysis using shorter time horizons and unit regression weights. The results are similar.

[Table 3]

4.2 Borrowers

We present our main results on the lending effects for the three episodes in Table 4. Results for PSI, OMT, and QE differ significantly. Exposure to sovereign debt has a strong positive effect for QE, but no detectable effect for PSI and OMT. Comparing two banks who are a standard deviation apart in terms of exposure to affected securities, average lending to firms by 34% for QE, by 10% for PSI, and by 6% for OMT (Columns 1 to 3 of Table 4). However, we cannot reject the null that the effect of the PSI and OMT interventions is zero at conventional levels of significance. For households, we find similar patterns with lending increasing by 11%, 4%, and 1% respectively (Columns 3 to 6 of Table 4). In Table A5, we repeat the analysis for monthly and quarterly horizons. In Table A6, we repeat the analysis for a closed sample, i.e. excluding borrower-bank relationships that do not exist in at least one event. We get similar results.

[Table 4]

In Table 5, rates of relationship termination follow similar patterns. Exit rates for firms decrease by 4 p.p., 1 p.p. and 0.5 p.p. for QE, PSI, and OMT respectively (Columns 1 to 3 of Table 5). For households, we get a decrease in the exit rates of 2 p.p. and 0.5 p.p. for QE and PSI and a increase in the exit rate of 0.04 p.p. for OMT (Columns 3 to 6 of Table 5), with the latter two estimates not statistically significantly different from zero. In Table A7, we repeat the analysis for shorter time horizons. We also repeat the analysis for relationship terminations without including borrower fixed effect from Equation 4. For firms, we add the following controls: log firm assets, log total workers, EBITDA/assets, equity ratio and an indicator variable that is equal to 1 if the firm has 90-day overdue loans. We report coefficients in Table A8.

[Table 5]

We now turn from the effects on firms and households that have no existing relationship with the bank to the formation of new relationships. We find a similar pattern of statistically and economically large effects for QE, with moderate to no effects for PSI and OMT (see Table 6). The acceptance rates for firms increases by 4% for QE and by 2% for PSI. We observe no significant effect for OMT. Households do not experience statistically significant changes in the aftermath of any event. In Table A9, we repeat the analysis for different time horizons.

We include borrower fixed effects in the main model. As most borrowers may apply only to one bank, the treated group might be highly selected, particularly in the case of households. Alternatively, we estimate Equation 4 excluding the borrower fixed effect θ_i . For firms, we add the following controls: log firm assets, log total workers, EBITDA/assets, equity ratio and an indicator variable that is equal to 1 if the firm has 90-day overdue loans. We report coefficients in Table A10. Results are directionally similar but statistically weaker.

[Table 6]

Lending and creditor composition We analyze the relationship between exposure to government debt securities and creditor composition in Table 7. The Bank of Portugal computes probabilities of default for firms (Antunes et al. (2016)). We restrict the sample to firms with available probability of default estimates. In Columns 1 to 3 the dependent variable is the 1-year change in log loans. In Columns 4 to 6 the dependent variable is an indicator equal to 1 for successful loan applications. We find some evidence that lenders exposed to government bonds increased their relative exposure to riskier borrowers after QE. In the intensive margin there is a negative relationship between the effect of exposure to government debt securities on lending and firm risk after PSI and OMT. In the extensive margin, exposed banks are more likely to establish new credit relationships with riskier borrowers after QE.

[Table 7]

The role of bank net worth on lending In Table 8 we report coefficients from estimating Equation 5. The elasticity of lending with respect to price appreciation is negative for firms and not statistically significant from 0 for households. However, we find a positive and significant elasticity of lending with respect to capital gains, particularly for QE.

In Table A11 we estimate the relationship between loan growth and capital gains using the instrumental variables model from Equation 7. This exercise has illustrative purposes only, as banks do not choose their exposure to sovereign debt randomly (the independence assumption of the instrumental variables model does not necessarily hold). Again, we find a positive relationship between capital gains and loan growth, particularly after the QE event. These results suggest that incentivizing banks to sell securities and realize accumulate capital gains stimulates the bank lending channel than increasing the price of securities held by banks.

[Table 8]

Pre-trends In Equation 4, we control for the effect of demand by including borrower fixed effects on lending. However, the effect of each event on lending could be driven by other factors that were previously affecting borrower-bank relationships. In Table A12, we re-estimate Equation 4 using pre-trends variables. In Table A12a, the dependent variable is delta log lending between one year before the event and the last month before the event. In Table A12b, the dependent variable is an indicator variable that is equal to 1 for borrower-bank relationships that do not exist 1 year before the event. We observe a positive and strong relationship between lending and exposure for PSI but not for other events.

In Table A13 we estimate Equation 4 again but include the pre-trend variables from Table A12 as additional controls. In Table A13a we estimate the effect of exposure on lending and include the change in lending in the year before the event as an additional control variable. In Table A13b, we estimate the effect of exposure on the probability of relationship termination using an indicator variable that is equal to 1 for borrower-bank relationships that do not exist 1 year before the event as an additional control. We find that coefficients from Table A13 are very similar to coefficients reported in Tables 4 and 5.

4.3 Firm Characteristics

We analyze the effect of exposure to government debt securities on real outcomes by aggregating outcomes at the firm level. We aggregate lending outcomes from the credit register at the firm level. Additionally, we retrieve data on average interest rates, sales and capital expenditure (CAPEX) from firm financial statements.

We measure the effect of exposure to government debt securities on two group of firms. First, we analyze firms that had pre-existing credit relationships. Our definition of firm-level exposure for existing credit relationships is similar to the one used by Jasova et al. (2021). Exposure is given by:

$$exposure_i = \frac{\sum_b (exposure_b \times credit_{bi})}{\sum_b (credit_{bi})} \tag{9}$$

where $exposure_i$ is the firm-level exposure, $exposure_b$ is the bank-level exposure variable used in the rest of the analysis, and $credit_{bi}$ is the outstanding amount of the firm-bank credit relationship. In the extensive margin we define exposure as:

$$exposure_i = \frac{\sum_b (exposure_b)}{B_i} \tag{10}$$

where B_i is the number of loan applications between the month of the event and one year after

the event.

In Table 9 we estimate Equation 8 using firm-level outcomes as the dependent variable. Table 9a depicts coefficients for firms with existing credit relationships (intensive margin). Table 9b depicts coefficients for firms that apply for new loans in the year after ech event (extensive margin). Consistent with the findings of Table 4, a one standard deviation increase in exposure for firms with existing credit relationships increases lending by 11% with QE, but only increases lending by approximately 3.5% with PSI and OMT. We do not find a strong relationship between exposure and interest rates for firms. We observe the same pattern in the extensive margin. A one standard deviation increase in exposure to government debt securities increases lending by 9.5% with QE, but only increases lending by 2.7% with PSI and has no statistically significant effect with OMT. The relationship between exposure to government debt securities and interest rates is economically small and statistically not very strong, as coefficients (measured in percentage points) are always smaller than 0.1 and mostly not significant at commonly reported significance levels (10%, 5% and 1%).

[Table 9]

We have fewer observations in the interest rate regressions than in the lending regressions. This difference is caused by two factors. First, some firms may have no loans. In these cases it is impossible to get an interest rate. Second, we use firm financial statements to obtain interest rates, and firms may not report financial statements in some years.

In Table 10 we measure the effect of exposure to government debt securities on sales and on the ratio of capital expenditure (CAPEX) to sales. We find evidence of a small but positive effect of exposure on the investment rate with PSI and OMT⁴ and find no statistically significant effect with QE.

[Table 10]

In the extensive margin we find that a one standard deviation increase in exposure to government debt securities is associated with a growth in sales of 3% with QE. However, CAPEX grows more

 $^{^4\}mathrm{The}$ average CAPEX/sales ratio is 11.1% for firms in the intensive margin sample and 9.4% for firms in the extensive margin sample.

slowly than sales with QE, as a one standard deviation increase in exposure is associated with a 0.7 p.p. decrease in the CAPEX/sales ratio. We also find a negative relationship between exposure and CAPEX/sales for PSI, but this relationship is weaker.

5 Conclusion

In this paper we make a comprehensive analysis of the relationship between the balance sheet channel and lending using data from Portugal. The Portuguese case is ideal to study balance sheet shocks. First, Portugal was affected by three large shocks – the Greek debt restructuring (PSI), Outright Monetary Transactions (OMT), and Quantitative Easing (QE). The the case of QE, the central bank actively intervened in banks' balance sheets. In the other cases, banks were only affected through a positive effect net worth shock. Second, Portugal has comprehensive credit and security registers covering both firms and households, and monthly bank balance sheet data. The richness of the data allows us to study these shocks in a unified framework.

We show that PSI and OMT had little impact on bank balance sheets. The price of government securities increased after these events but banks did not reduce their government debt holdings. QE had a smaller impact on the value of securities held by banks. However, banks that were more exposed to sovereign debt reduced their holdings and realized previously accumulated capital gains.

We analyze the transmission of each shock to firms and households through the bank balance sheet channel. Exposure to sovereign debt securities had a moderate to null effect on lending for PSI and OMT and a largely positive effect for QE, both in the intensive and extensive margins. We also find a positive relationship between exposure to sovereign debt and firm-level real outcomes for QE, particularly in the case of new loan applicants. Additionally, we observe a positive relationship between realized capital gains and lending after QE. However, we do not find a positive relationship between increases in security prices and lending.

Summing up, the main finding of this paper is that bank balance sheet shocks were not all created equal. The calibration of quantitative macro-finance models should consider that the balance sheet channel is considerably stronger when there is the provision of liquidity to banks' balance sheets.

Tables

Table 1: Descriptive Statistics

	Weig	ghted	Unweighted		
	Mean (1)	Std. dev. (2)	Mean (3)	Std. dev. (4)	
% exposed assets	4.25	3.92	2.29	6.04	
Total assets (EUR M)	$51,\!655.41$	$45,\!297.31$	$10,\!678.41$	$26,\!384.53$	
Capital ratio	14.12	12.90	18.83	16.91	
% arrears (firms)	10.49	8.89	8.98	11.31	
% arrears (households)	5.61	7.11	6.83	12.38	
% accepted applications (firms)	12.86	11.45	11.99	19.57	
% accepted applications (households)	21.89	15.94	11.82	17.50	
% LTRO	0.45	0.75	0.15	1.60	
Observations	132		132		

(a) Bank Descriptive Statistics

(b) Borrower	Descriptive	Statistics
--------------	-------------	------------

	Fi	irms	Households	
	Mean (1)	Std. dev. (2)	Mean (3)	Std. dev. (4)
Δ log loans (1 month)	-0.28	1.44	-0.32	1.25
Δ log loans (3 months)	-0.54	1.98	-0.57	1.69
Δ log loans (12 months)	-1.51	3.24	-1.33	2.58
Exit rate (1 month)	3.11	17.35	4.96	21.71
Exit rate (3 months)	5.77	23.32	8.63	28.07
Exit rate (12 months)	15.24	35.94	18.97	39.21
Relationship length (years)	4.56	2.83	3.1	1.62
Observations	684	4,426	679	9,299

Notes. The table shows descriptive statistics for banks, firm-bank and household-bank pairs used in the analysis. In Columns (1) and (2) of Table 1a, statistics are weighted by the total number of customers (firms and households). In Columns (3) and (4) of Table 1a we use sample weights. Relationship length for households is censored to the left because the household credit register only starts in 2009. We use the month before each event as the reference date: January 2012 for PSI, June 2012 for OMT, and December 2014 for QE.

		$\frac{\Delta \text{ Exposed securities}}{\text{Total assets}}$	
	$\begin{array}{c} \mathrm{PSI} \\ (1) \end{array}$	$\begin{array}{c} \text{OMT} \\ (2) \end{array}$	$\begin{array}{c} ext{QE} \\ ext{(3)} \end{array}$
Exposure	0.001	-0.003	-0.019**
	(0.002)	(0.003)	(0.009)
Bank controls	Yes	Yes	Yes
Observations	44	44	44
R-squared	0.209	0.296	0.616

Table 2: Regression Results: Security Holdings

Notes. The table reports estimates of coefficient β_1 from Equation 2. The dependent variable $\frac{\Delta \text{ Exposed securities}}{\text{Total assets}}$ is the change in the ratio of exposed securities over total assets one year after the event. The independent variable Exposure is the standardized ratio of exposed securities to total bank assets in the last month before the event. We use bank-level characteristics from Table A1 as additional controls. Observations are weighted by the number of bank customers (firms and households) in the sample. Robust standard errors are reported in parentheses. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

	Δ log net worth						
	Pri	ce apprecia	tion	(Capital gain	ıs	
	$\begin{array}{c} \text{PSI} \\ (1) \end{array}$	OMT (2)	$\begin{array}{c} \text{QE} \\ (3) \end{array}$	PSI (4)	$\begin{array}{c} \text{OMT} \\ (5) \end{array}$	QE (6)	
Exposure	$\begin{array}{c} 0.182^{**} \\ (0.071) \end{array}$	$0.062 \\ (0.049)$	$0.027 \\ (0.025)$	0.001 (0.073)	-0.012 (0.053)	0.024^{***} (0.006)	
Bank controls Observations R-squared	Yes 44 0.442	Yes 44 0.501	Yes 44 0.510	Yes 44 0.330	Yes 44 0.424	Yes 44 0.618	

Table 3: Regression Results: Bank-level Net Worth Gains

Notes. The table reports estimates of coefficient β_1 from Equation 2. The dependent variable Δ log net worth is the change in log bank net worth induced by price appreciation (Columns 1 to 3) and capital gains (Columns 4 to 6) one year after the event. We obtain gains from price appreciation by multiplying monthly price variation by outstanding security holdings. Realized gains are cumulative gains that banks monetize when they close their position in a certain security. We compute the effect of price variation and capital gains on bank net worth using Equation 1. The independent variable Exposure is the standardized ratio of exposed securities to total bank assets in the last month before the event. We use bank-level characteristics from Table A1 as additional controls. Observations are weighted by the number of bank customers (firms and households) in the sample. Robust standard errors are reported in parentheses. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

	Δ log outstanding loans								
_		Firms			Households				
$\begin{array}{c ccc} PSI & OMT & QE \\ \hline (1) & (2) & (3) \end{array}$		PSI OMT QE (1) (2) (3)		PSI (4)	$\begin{array}{c} \text{OMT} \\ (5) \end{array}$	$\begin{array}{c} ext{QE} \\ ext{(6)} \end{array}$			
Exposure	0.109 (0.083)	0.064 (0.090)	0.336^{***} (0.084)	$0.042 \\ (0.036)$	0.013 (0.035)	0.110^{***} (0.018)			
Observations R-squared	$241,311 \\ 0.433$	$233,402 \\ 0.434$	$209,713 \\ 0.426$	$242,823 \\ 0.461$	$231,\!840 \\ 0.468$	$204,\!636 \\ 0.469$			

Table 4: Regression Results: Outstanding Loans

Notes. The table shows coefficients from estimating Equation 4. The dependent variable Δ log outstanding loans is the change in one plus log outstanding loans one year after the event. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. We control for bank characteristics with variables from Table 1. Errors are clustered at bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

	Relationship ends $= 1$								
_		Firms							
_	$\begin{array}{c ccc} \hline PSI & OMT & QE \\ (1) & (2) & (3) \end{array}$			PSI (4)	$\begin{array}{c} \text{OMT} \\ (5) \end{array}$	QE (6)			
Exposure	-0.010 (0.008)	-0.005 (0.009)	-0.039^{***} (0.008)	-0.003 (0.006)	0.0004 (0.005)	-0.019^{***} (0.002)			
Observations R-squared	$241,311 \\ 0.427$	$233,402 \\ 0.43$	$209,713 \\ 0.43$	$242,823 \\ 0.4667$	$231,\!840 \\ 0.4741$	$204,636 \\ 0.477$			

Table 5: Regression Results: Termination of Credit Relationships

Notes. The table shows coefficients from estimating Equation 4. The dependent variable Relationship ends = 1 is an indicator variable that is equal to 1 if the borrower does not have a credit relationship with the bank 1 year after the event. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. We control for bank characteristics with variables from Table 1. Errors are clustered at bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

	Accepted application $= 1$								
		Firms			Households				
_	PSI (1)	$\begin{array}{c} \text{OMT} \\ (2) \end{array}$	QE (3)	PSI (4)	$\begin{array}{c} \text{OMT} \\ (5) \end{array}$	QE (6)			
Exposure	0.023^{**} (0.009)	-0.001 (0.010)	0.036^{***} (0.007)	-0.002 (0.006)	-0.011 (0.009)	0.010 (0.011)			
Observations R-squared	$14,676 \\ 0.45$	$16,279 \\ 0.459$	$19,861 \\ 0.456$	$13,\!051 \\ 0.456$	$13,477 \\ 0.454$	$16,542 \\ 0.461$			

 Table 6: Regression Results: Loan Applications

Notes. The table shows coefficients from estimating Equation 4. The dependent variable Accepted application = 1 is an indicator variable that is equal to 1 if the loan application is accepted. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. The sample consists of loan applications between the 4th and the 12th month after the month of the event. We control for bank characteristics with variables from Table 1. Errors are clustered at bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

	$\Delta \log$	outstanding	; loans	Accept	Accepted application $= 1$		
	$\begin{array}{c} \mathbf{PSI} \\ (1) \end{array}$	$\begin{array}{c} \text{OMT} \\ (2) \end{array}$	QE (3)	$\begin{array}{c} \mathrm{PSI} \\ (4) \end{array}$	$\begin{array}{c} \text{OMT} \\ (5) \end{array}$	$\begin{array}{c} ext{QE} \\ ext{(6)} \end{array}$	
Exposure	0.176^{**} (0.084)	0.113 (0.099)	$\begin{array}{c} 0.249^{***} \\ (0.059) \end{array}$	0.018^{**} (0.008)	-0.007 (0.011)	0.030^{**} (0.011)	
$Exposure \times P(default)$	-0.005^{*} (0.002)	-0.005** (0.003)	0.001 (0.006)	$0.106 \\ (0.102)$	$0.142 \\ (0.098)$	0.122^{**} (0.055)	
Observations R-squared	$203,285 \\ 0.436$	$197,405 \\ 0.438$	$164,066 \\ 0.437$	$14,\!051 \\ 0.45$	$15,\!676 \\ 0.46$	$18,002 \\ 0.451$	

Table 7: Regression Results: By Borrower Composition

Notes. The table shows coefficients from estimating Equation 4. In Columns 1 to 3 the dependent variable Δ log outstanding loans is the change in log one plus outstanding loans one year after the event. In Columns 4 to 6 the dependent variable Accepted application = 1 is an indicator equal to 1 for firms with accepted loan applications 4-12 months after the event. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. The independent variable Exposure \times P(default) is the interaction between Exposure and the probability of default provided by Antunes et al. (2016). We control for bank characteristics with variables from Table 1. Errors are clustered at bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

	Δ log outstanding loans						
		Firms			Households		
N7 • 11	PSI	OMT	QE	PSI	OMT	QE	
Variable	(1)	(2)	(3)	(4)	(5)	(6)	
Price appreciation	-0.764^{**}	-1.877^{***}	-2.873	-0.014	1.794	0.367	
	(0.339)	(0.011)	(2.001)	(0.294)	(1.100)	(0.007)	
Capital gains	-0.288 (0.291)	$1.524 \\ (1.191)$	$\begin{array}{c} 4.589^{**} \\ (2.200) \end{array}$	-0.061 (0.217)	-3.085^{*} (1.577)	$0.498 \\ (1.509)$	
Observations R-squared	$266,307 \\ 0.417$	$258,829 \\ 0.418$	$235,723 \\ 0.400$	$244,735 \\ 0.460$	$233,\!642 \\ 0.467$	$206,515 \\ 0.468$	

Table 8: Regression Results: Lending, Price Appreciation and Capital Gains

Notes. The table shows coefficients from estimating Equation 4. The dependent variable Δ log outstanding loans is the change in one plus log outstanding loans one year after the event. The independent variable Price appreciation is the change in log bank net worth induced by price appreciation. Capital gains is the change in log bank net worth induced by realized capital gains. The independent variables are defined with more detail in the text. We control for bank characteristics with variables from Table 1. Errors are clustered at the bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

	Log	outstanding l	oans		Interest Rate	
-	PSI	OMT	QE	PSI	OMT	QE
Variable	(1)	(2)	(3)	(4)	(5)	(6)
Exposure	0.035^{***}	0.036^{***}	0.113***	-0.010	0.035	-0.034
	(0.009)	(0.009)	(0.009)	(0.022)	(0.022)	(0.026)
	· /			× ,		
Observations	1,396,656	1,372,482	1,301,640	923,186	901,385	665,307
R-squared	0.831	0.833	0.849	0.734	0.735	0.784
		(b)) Loan Applicat	tions		
	Log	outstanding l	oans		Interest rate	
_	PSI	OMT	QE	PSI	OMT	QE
Variable	(1)	(2)	(3)	(4)	(5)	(6)
Exposure	0.027**	-0.007	0.095^{***}	0.055^{*}	-0.060*	-0.013
	(0.012)	(0.012)	(0.011)	(0.029)	(0.032)	(0.030)
	. ,	. ,	. ,	. ,	. ,	. ,
Observations	745,668	735,438	1,040,262	498,966	490,775	437,262
R-squared	0.913	0.914	0.907	0.728	0.728	0.786

Table 9: Regression Results: Firm-level Outcomes (Lending)

(a) Existing Credit Relationships

Notes. The table shows coefficients from estimating Equation 8. Table 9a depicts coefficients for firms with existing credit relationships in the month before the event. Table 9b depicts coefficients for firms that applied for loans in the year after the event. The dependent variable Log outstanding loans is the change in log outstanding loans one year after the event. The dependent variable Interest Rate is the average interest rate paid by firms. Exposure is given by Equations 9 and 10. Errors are clustered at the firm level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

			-			
		Log Sales			CAPEX/Sale	s
_	PSI	OMT	QE	PSI	OMT	QE
Variable	(1)	(2)	(3)	(4)	(5)	(6)
Exposure	-0.003	0.014^{*}	-0.006	0.003^{**}	0.003^{**}	0.0003
	(0.008)	(0.008)	(0.008)	(0.002)	(0.002)	(0.002)
Observations	1,072,841	1,046,565	$933,\!810$	980,886	$958,\!210$	$846{,}578$
R-squared	0.866	0.867	0.897	0.628	0.634	0.638
		(b)) Loan Applica	tions		
		Log Sales			CAPEX/Sale	s
_	PSI	OMT	QE	PSI	OMT	QE
Variable	(1)	(2)	(3)	(4)	(5)	(6)
Exposure	0.003	-0.004	0.030^{***}	-0.004**	-0.002	-0.007***
	(0.009)	(0.009)	(0.008)	(0.002)	(0.002)	(0.002)
Observations	$617,\!178$	$607,\!964$	$771,\!656$	$566,\!540$	$558,\!472$	$694,\!137$
R-squared	0.888	0.893	0.905	0.599	0.597	0.684

Table 10: Regression Results: Firm-level Outcomes (Sales and Investment)

(a) Existing Credit Relationships

Notes. The table shows coefficients from estimating Equation 8. Table 10a depicts coefficients for firms with existing credit relationships in the month before the event. Table 10b depicts coefficients for firms that applied for loans in the year after the event. The dependent variable Log Sales is the change in log sales one year after the event. The dependent variable CAPEX/Sales is the ratio of capital expenditure to sales. Exposure is given by Equations 9 and 10. Errors are clustered at the firm level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

Figures



Figure 1: Government Bond Yields of Periphery Countries

Notes. The table shows 10-year government bond yields for Greece, Portugal, Ireland, Italy and Spain.



Figure 2: Exposure to Sovereign Debt

Notes. The figure shows the book value of sovereign debt securities held by banks in the sample as a percentage of total assets. We divide securities by issuer country. *Portugal* includes securities issued by the Portuguese government. *Other periphery countries* includes securities issued by Greece, Spain, Italy and Ireland. *Other Euro Area countries* includes securities issued by the remaining countries from the Euro Area.





Notes. The figure reports coefficients and 95% confidence intervals from Equation 3. The dependent variable is the cumulative variation in exposed security holdings as a percentage of total assets in the month before the Quantitative Easing event. The independent variable is the standardized ratio of exposed securities to total assets, measured in the last month before the event.





Notes. The figure shows the percentage of ECB Portuguese government debt security purchases whose counterparty was a bank in the sample, weighted by security market value.

References

- ABBASSI, P., R. IYER, J.-L. PEYDRÓ, AND F. R. TOUS (2016): "Securities trading by banks and credit supply: Micro-evidence from the crisis," *Journal of Financial Economics*, 121, 569 – 594.
- ACHARYA, V. V., T. EISERT, C. EUFINGER, AND C. HIRSCH (2017): "Whatever it takes: The Real Effects of Unconventional Monetary Policy," *Working Paper*.
- ACHARYA, V. V. AND S. STEFFEN (2015): "The Greatest Carry Trade Ever? Understanding Eurozone Bank Risks," *Journal of Financial Economics*, 115, 215–236.
- ANTUNES, A., H. GONÇALVES, P. PREGO, ET AL. (2016): "Firm default probabilities revisited," Economic Bulletin and Financial Stability Report Articles.
- B GROSSE-RUESCHKAMP, S STEFFEN, D STREITZ (2017): "Cutting Out the Middleman–The ECB as Corporate Bond Investor," *Working Paper*.
- BECKER, B. AND V. IVASHINA (2017): "Financial Repression in the European Sovereign Debt Crisis*," *Review of Finance*, 22, 83–115.
- BERTRAND, M., E. DUFLO, AND S. MULLAINATHAN (2004): "How much should we trust differences-in-differences estimates?" *The Quarterly Journal of Economics*, 119, 249–275.
- BOTTERO, M., S. LENZU, AND F. MEZZANOTTI (2019): "Sovereign Debt Exposure and Bank Lending Channel: Impact on Credit Supply and the Real Economy," *Journal of International Economics*.
- BRUNNERMEIER, M. K. AND Y. SANNIKOV (2016): "The I theory of money," Working Paper.
- CARPINELLI, L. AND M. CROSIGNANI (2017): "The effect of central bank liquidity injections on bank credit supply," *Working Paper*.
- CHAKRABORTY, I., I. GOLDSTEIN, AND A. MACKINLAY (2017): "Monetary Stimulus and Bank Lending," *Working Paper*.

- DELL'ARICCIA, G., P. RABANAL, AND D. SANDRI (2018): "Unconventional Monetary Policies in the Euro Area, Japan, and the United Kingdom," *Journal of Economic Perspectives*, 32, 147–72.
- DI MAGGIO, M., M. A. KERMANI, AND C. PALMER (2016): "How Quantitative Easing Works: Evidence on the Refinancing Channel," *Working Paper*.
- GOLDSTEIN, I., J. WITMER, AND J. YANG (2018): "Following the money: Evidence for the portfolio balance channel of quantitative easing," *Working Paper*.
- HOLMSTROM, B. AND J. TIROLE (1997): "Financial intermediation, loanable funds, and the real sector," *Quarterly Journal of Economics*, 112, 663–691.
- JASOVA, M., C. MENDICINO, AND D. SUPERA (2021): "Policy uncertainty, lender of last resort and the real economy," *Journal of Monetary Economics*, 118, 381–398.
- JIMÉNEZ, G., S. ONGENA, J.-L. PEYDRÓ, AND J. SAURINA (2012): "Credit supply and monetary policy: Identifying the bank balance-sheet channel with loan applications," *American Economic Review*, 102, 2301–26.
- KASHYAP, A. AND J. STEIN (1994): "Monetary policy and bank lending," *Monetary Policy*, 221–256.
- KHWAJA, A. I. AND A. MIAN (2008): "Tracing the impact of bank liquidity shocks: Evidence from an emerging market," *American Economic Review*, 98, 1413–42.
- KOIJEN, R. S., F. KOULISCHER, B. NGUYEN, AND M. YOGO (2017a): "Euro-area quantitative easing and portfolio rebalancing," *American Economic Review*, 107, 621–27.
- KOIJEN, R. S. J., F. KOULISCHER, B. NGUYEN, AND M. YOGO (2017b): "Euro-Area Quantitative Easing and Portfolio Rebalancing," *American Economic Review*, 107, 621–27.
- KRISHNAMURTHY, A., S. NAGEL, AND A. VISSING-JORGENSEN (2018): "ECB policies involving government bond purchases: Impact and channels," *Review of Finance*, 22, 1–44.

- LUCK, S. AND T. ZIMMERMANN (2017): "Employment effects of unconventional monetary policy: Evidence from QE," *Working Paper*.
- MINK, M. AND J. DE HAAN (2013): "Contagion during the Greek sovereign debt crisis," *Journal* of International Money and Finance, 34, 102–113.
- MORAIS, B., J.-L. PEYDRÓ, J. ROLDÁN-PEÑA, AND C. RUIZ-ORTEGA (2019): "The International Bank Lending Channel of Monetary Policy Rates and QE: Credit Supply, Reachfor-Yield, and Real Effects," *The Journal of Finance*, 74, 55–90.
- ONGENA, S., A. POPOV, AND N. VAN HOREN (2019): "The Invisible Hand of the Government: Moral Suasion during the European Sovereign Debt Crisis," *American Economic Journal: Macroe*conomics, 11, 346–79.
- RODNYANSKY, A. AND O. DARMOUNI (2017): "The Effects of Quantitative Easing," *Review of Financial Studies*, 30, 3858–3887.
- STROEBEL, J. AND J. B. TAYLOR (2012): "Estimated impact of the Federal Reserve's mortgagebacked securities purchase program," *International Journal of Central Banking*, 8, 1–42.
- ZETTELMEYER, J., C. TREBESCH, AND M. GULATI (2013): "The Greek debt restructuring: an autopsy," *Economic Policy*, 28, 513–563.

Appendix Tables

Table A1: Descriptive Statistics by Event (PSI)

	Weig	ghted	Unwe	ighted
	Mean (1)	Std. dev.	Mean (3)	Std. dev.
 % exposed assets Total assets (EUR M) Capital ratio % arrears (firms) % arrears (households) % accepted applications (firms) % accepted applications (households) % LTRO 	$\begin{array}{c} 3.50 \\ 54,972.23 \\ 12.22 \\ 8.02 \\ 5.77 \\ 13.11 \\ 20.60 \\ 0.31 \end{array}$	$\begin{array}{c} 3.34\\ 48,194.50\\ 12.47\\ 6.83\\ 7.13\\ 11.78\\ 13.69\\ 0.51\end{array}$	$1.78 \\ 11,319.20 \\ 18.04 \\ 7.09 \\ 6.31 \\ 14.68 \\ 10.75 \\ 0.23$	$\begin{array}{c} 4.03\\ 28,330.86\\ 17.49\\ 8.66\\ 11.63\\ 21.40\\ 14.47\\ 1.09\end{array}$
Observations (b) Bor	4 rrower Descrip	14 ptive Statistics	4	4
	Fii	rms	House	eholds
	Mean (1)	Std. dev. (2)	$\begin{array}{c} \text{Mean} \\ (3) \end{array}$	Std. dev. (4)
Δ log loans (Month) Δ log loans (Quarter) Δ log loans (Year) Exit rate (Month)	-0.29 -0.55 -1.55 3.19	1.44 1.97 3.21 17.58	-0.33 -0.58 -1.36 5.04	1.28 1.71 2.58 21.88
Exit rate (Quarter) Exit rate (Year) Relationship length (years)	5.87 15.46 4.08	$23.51 \\ 36.16 \\ 2.37$	8.76 19.36 2.45	$28.27 \\ 39.51 \\ 0.92$
Observations	241	,311	241	,311

(a) Bank Descriptive Statistics

Table A1: Descriptive Statistics by Event (OMT)

	Weig	shted	Unwei	ighted
-	Mean (1)	Std. dev. (2)	Mean (3)	Std. dev. (4)
 % exposed assets Total assets (EUR M) Capital ratio % arrears (firms) % arrears (households) % accepted applications (firms) % accepted applications (households) % LTRO 	$\begin{array}{c} 4.26\\ 55,261.49\\ 13.43\\ 9.96\\ 5.43\\ 13.10\\ 24.37\\ 0.37\end{array}$	$\begin{array}{r} 4.05\\ 48,491.42\\ 13.27\\ 8.49\\ 7.07\\ 12.15\\ 17.85\\ 0.86\end{array}$	$\begin{array}{c} 2.27\\ 11,339.60\\ 19.12\\ 8.30\\ 6.70\\ 11.98\\ 15.26\\ 0.14 \end{array}$	$5.48 \\28,488.66 \\18.08 \\9.97 \\13.06 \\19.93 \\21.82 \\0.55$
Observations	4	4	4	4
(d) Bor	rower Descrip	otive Statistics		

(c) Bank Descriptive Statistics

	F	irms	Hous	seholds
	Mean	Std. dev.	Mean	Std. dev.
	(1)	(2)	(3)	(4)
$\Delta \log \text{ loans (Month)}$	-0.3	1.44	-0.32	1.23
$\Delta \log \text{ loans (Quarter)}$	-0.57	1.98	-0.57	1.68
$\Delta \log \log (Year)$	-1.5	3.2	-1.33	2.56
Exit rate (Month)	3.22	17.64	5.02	21.83
Exit rate (Quarter)	5.99	23.73	8.78	28.31
Exit rate (Year)	15.12	35.82	19.11	39.31
Relationship length (years)	4.32	2.49	2.76	1.06
Observations		3 402		3 402

Table A1: Descriptive Statistics by Event (QE)

(\mathbf{e}	Bank	Descriptive	Statistics
---	--------------	------	-------------	------------

	Weig	chted	Unwe	ighted
-	Mean	Std. dev.	Mean	Std. dev.
	(1)	(2)	(3)	(4)
% exposed assets	5.06	4.20	2.29	6.04
Total assets (EUR M)	$44,\!168.39$	36,783.05	$10,\!678.41$	$26,\!384.53$
Capital ratio	16.93	12.48	18.83	16.91
% arrears (firms)	13.76	10.22	8.98	11.31
% arrears (households)	5.64	7.13	6.83	12.38
% accepted applications (firms)	12.35	10.23	11.99	19.57
% accepted applications (households)	20.66	15.77	11.82	17.50
% LTRO	0.69	0.79	0.15	1.60
Observations	4	4	4	4
(f) Bor	rower Descrip	tive Statistics		
	Fir	rms	House	eholds
-	Mean	Std. dev.	Mean	Std. dev.
	(1)	(2)	(3)	(4)
Δ log loans (Month)	-0.25	1.44	-0.32	1.23
Δ log loans (Quarter)	-0.48	2	-0.55	1.66
Δ log loans (Year)	-1.45	3.33	-1.28	2.61
Exit rate (Month)	2.89	16.74	4.79	21.36
Exit rate (Quarter)	5.41	22.62	8.29	27.57
Exit rate (Year)	15.12	35.83	18.35	38.71
Relationship length (years)	5.37	3.42	4.25	2.11
Observations	233	,402	233	,402

Notes. The table shows descriptive statistics for banks, firm-bank and household-bank pairs used in the analysis. In Columns (1) and (2) of Tables A1a, A1c, and A1e, statistics are weighted by the total number of customers (firms and households) in the analysis. In Columns (3) and (4) of Tables A1a, A1c, and A1e, we use sample weights. Relationship length for households is censored to the left because the household credit register only starts in 2009. We use the month before each event as the reference date: January 2012 for PSI, June 2012 for OMT, and December 2014 for QE.

(a)	Weighted
	(a)

 Table A2:
 Robustness:
 Bank-level
 Exposed
 Security
 Holdings

					A Exposed securiti Total assets	les			
		ISI			OMT			QE	
	Month (1)	Quarter (2)	Year (3)	Month (4)	$\substack{\text{Quarter}\\(5)}$	Year (6)	Month (7)	Quarter (8)	$_{(9)}^{\rm Year}$
Exposure	0.001 (0.001)	0.002^{*} (0.001)	0.001 (0.002)	0.001 (0.0004)	-0.0004 (0.002)	-0.003 (0.003)	-0.007*(0.004)	-0.011^{***} (0.003)	-0.020^{**} (0.009)
Observations R-squared	$44 \\ 0.359$	44 0.392	44 0.209	44 0.267	$44 \\ 0.216$	44 0.296	$\frac{44}{0.411}$	$\frac{44}{0.595}$	$44 \\ 0.616$
				(b) Unv	weighted				
					A Exposed securiti Total assets	les			
		ISI			OMT			QE	
	Month (1)	Quarter (2)	Year (3)	Month (4)	$\operatorname{Quarter}_{(5)}$	Year (6)	Month (7)	Quarter (8)	$\substack{\text{Year}\\(9)}$
Exposure	0.003^{***} (0.001)	0.003^{***} (0.001)	-0.010*(0.006)	0.001 (0.0004)	-0.009^{***} (0.003)	-0.013^{***} (0.003)	-0.029^{***} (0.004)	-0.033^{***} (0.003)	-0.020^{***} (0.001)
Observations R-squared	$44 \\ 0.515$	$44 \\ 0.589$	44 0.457	44 0.433	44 0.700	44 0.615	$44 \\ 0.891$	$\frac{44}{0.917}$	44 0.877
<i>Notes.</i> The table securities over to securities to total observations are observations are	reports esti tal assets one l bank assets weighted by re reported i	mates of coeffic e month, one qu i in the last mon the number of n parentheses.	Sient β_1 from Equation Equation (19) in the form one y at the evaluation of the	uation 2. The d ear after the even vent. We use bar (firms and houss e significance at	ependent varial nt. The indepen ık-level charact eholds) in the s the 1%, 5%, an	ble $\frac{\Delta \operatorname{Exposed sec}}{\operatorname{Total assec}}$ ndent variable Exercisitics from Tab eristics from Table and 10% levels, ress	urities is the cha ts sposure is the sta ple A1 as additio ? A2b, we use ur spectively.	nge in the ratio andardized ratio nal controls. In nit sample weigl	of exposed o of exposed Table A2a, nts. Robust

ECB Working Paper Series No 2607 / October 2021

Table .	A3:	Robustness:	Price Appreciation
		(a) Weigh	ited

Weighted	
(a)	

					∆ log net wort	h			
		ISd			OMT			QE	
	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)	Month (7)	Quarter (8)	Year (9)
Exposure	0.0866^{**} (0.0400)	0.110^{**} (0.0461)	0.182^{**} (0.0705)	0.0779^{*} (0.0449)	-0.00448 (0.0329)	0.0615 (0.0487)	0.00472^{***} (0.00134)	0.0332^{**} (0.0134)	0.0267 (0.0247)
Observations R-squared	44 0.378	44 0.410	$44 \\ 0.442$	$\frac{44}{0.472}$	$44 \\ 0.485$	$44 \\ 0.501$	$44 \\ 0.140$	$44 \\ 0.486$	$\frac{44}{0.510}$
				(b) Unv	veighted				
					∆ log net wort	h			
		ISd			OMT			QE	
	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)	Month (7)	Quarter (8)	Year (9)
Exposure	0.060* (0.031)	0.094^{**} (0.040)	0.19^{***} (0.063)	0.008 (0.018)	0.028 (0.019)	0.082^{***} (0.020)	0.032^{*} (0.018)	0.038** (0.018)	-0.032* (0.019)
Observations R-squared	$\frac{44}{0.197}$	$\frac{44}{0.257}$	44 0.397	$\frac{44}{0.157}$	$\begin{array}{c} 44 \\ 0.196 \end{array}$	44 0.329	$44 \\ 0.128$	$44 \\ 0.130$	$\frac{44}{0.140}$
<i>Notes.</i> The table by price apprecia securities to total observations are standard errors a	reports estin tion one mo bank assets weighted by re reported in	nates of coefficient nth, one quarte in the last mouther number of n parentheses.	ent β_1 from Equa sr, and one year ath before the ev bank customers ***, **, denote	ution 2. The dept after the event. ent. We use ban (firms and house s significance at t	endent variable The independ k-level charact sholds) in the s the 1%, 5%, an	Δ log net worth lent variable Exp eristics from Table sample. In Table d 10% levels, res	is the change in l posure is the stan ble A1 as addition ? A3b, we use uni spectively.	og bank net wo ıdardized ratio ıal controls. In it sample weigh	rth induced of exposed Table A3a, tts. Robust

Gains	
Capital	
Robustness:	
A4:	
Table	

Weighted
(a)

					∆ log net wort	h			
		PSI			OMT			QE	
	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)	Month (7)	Quarter (8)	Year (9)
Exposure	-0.004 (0.062)	0.002 (0.061)	0.001 (0.072)	0.002^{**} (0.001)	-0.077 (0.047)	-0.013 (0.053)	0.006^{*} (0.003)	0.014^{**} (0.005)	0.024^{***} (0.006)
Observations R-squared	$44 \\ 0.298$	$\frac{44}{0.333}$	$44 \\ 0.330$	$\begin{array}{c} 44 \\ 0.150 \end{array}$	$\begin{array}{c} 44 \\ 0.459 \end{array}$	$44 \\ 0.424$	$\frac{44}{0.275}$	$\frac{44}{0.528}$	$\frac{44}{0.618}$
				(b) Unv	veighted				
					∆ log net wort	h			
		ISI			OMT			QE	
	Month (1)	Quarter (2)	$\begin{array}{c} \text{Year} \\ (3) \end{array}$	Month (4)	Quarter (5)	Year (6)	Month (7)	Quarter (8)	Year (9)
Exposure	-0.016 (0.024)	-0.014 (0.024)	0.029 (0.027)	0.0004^{*} (0.0002)	-0.012 (0.015)	0.013 (0.019)	0.028 (0.018)	0.042^{**} (0.018)	0.037^{**} (0.018)
Observations R-squared	$44 \\ 0.124$	$\frac{44}{0.134}$	$44 \\ 0.150$	44 0.083	$44 \\ 0.051$	$44 \\ 0.135$	$44 \\ 0.053$	$\frac{44}{0.138}$	$\frac{44}{0.136}$
<i>Notes.</i> The table induced by capits securities to total observations are ³ standard errors au	e reports est al gains one bank assets weighted by re reported i	imates of coeffi month, one qua in the last mon the number of n parentheses.	cient β_1 from E rter, and one yei th before the ev bank customers ***, **, * denote	quation 2. The ar after the even ent. We use ban (firms and house s significance at t	dependent var t. The indepen k-level charact sholds) in the s the 1%, 5%, an	iable Δ log net ' ident variable Ex eristics from Tab sample. In Table d 10% levels, res	worth is the cha posure is the sta ole A1 as additio A4b, we use un spectively.	nge in log banl undardized ratic nal controls. In uit sample weigl	c net worth • of exposed Table A4a, nts. Robust

			$\Delta \log outsta$	anding loans		
_		Firms			Households	
	Month (1)	Quarter (2)	Year (3)	Month (4)	$\begin{array}{c} \text{Quarter} \\ (5) \end{array}$	Year (6)
Exposure	0.014 (0.013)	0.031 (0.024)	0.11 (0.083)	0.001 (0.013)	0.015 (0.022)	$0.042 \\ (0.036)$
Observations R-squared	$241,\!311 \\ 0.401$	$241,\!311\\0.411$	$241,311 \\ 0.433$	$242,823 \\ 0.435$	$242,823 \\ 0.441$	$242,823 \\ 0.461$
			(b) OMT			
			$\Delta \log outsta$	anding loans		
		Firms			Households	
_	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)
Exposure	0.004 (0.017)	0.025 (0.034)	0.064 (0.090)	$0.015 \\ (0.013)$	0.022 (0.020)	0.013 (0.036)
Observations R-squared	$233,402 \\ 0.404$	$233,402 \\ 0.411$	$233,402 \\ 0.434$	$231,\!840$ 0.453	$231,840 \\ 0.459$	$231,840 \\ 0.468$

Table A5: Robustness: Outstanding Loans (1/2)

(a) PSI

Table A5: Robustness: Outstanding Loans (2/2)

(c) QE

			$\Delta \log outst$	anding loans		
_		Firms			Households	
	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)
Exposure	0.051^{***} (0.010)	0.092^{***} (0.015)	0.34^{***} (0.084)	0.017^{*} (0.009)	0.026^{*} (0.014)	0.110^{***} (0.018)
Observations R-squared	$209,713 \\ 0.404$	$209,713 \\ 0.415$	$209,713 \\ 0.426$	$204,\!636 \\ 0.456$	$204,\!636 \\ 0.467$	$204,\!636 \\ 0.469$

Notes. The table shows coefficients from estimating Equation 4. The dependent variable Δ log outstanding loans is the change in one plus log outstanding loans one month, one quarter, and one year after the event. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. We control for bank characteristics with variables from Table 1. Errors are clustered at bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

			Δ log outsta	nding loans		
_		Firms			Households	
	Month (1)	Quarter (2)	Year (3)	Month (4)	$\begin{array}{c} \text{Quarter} \\ (5) \end{array}$	Year (6)
Exposure	0.0089^{**} (0.003)	0.015^{***} (0.004)	0.034^{**} (0.013)	-0.005 (0.005)	-0.002 (0.004)	-0.021 (0.013)
Observations R-squared	$130,351 \\ 0.395$	$130,\!351 \\ 0.405$	$130,351 \\ 0.405$	$108,939 \\ 0.457$	$\begin{array}{c} 108,\!939 \\ 0.458 \end{array}$	$108,939 \\ 0.47$
			(b) OMT			
			Δ log outsta	nding loans		
_		Firms			Households	
_	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)
Exposure	$0.005 \\ (0.003)$	0.017^{***} (0.005)	0.019 (0.015)	-0.007 (0.008)	-0.016 (0.011)	-0.030^{***} (0.010)
Observations R-squared	$130,351 \\ 0.411$	$130,351 \\ 0.408$	$130,351 \\ 0.41$	$108,939 \\ 0.476$	$108,939 \\ 0.479$	$108,939 \\ 0.469$

Table A6: Robustness: Outstanding Loans (Closed Sample) (1/2)

(a) PSI

			$\Delta \log outst$	anding loans		
_		Firms			Households	
	Month (1)	Quarter (2)	Year (3)	Month (4)	$\begin{array}{c} \text{Quarter} \\ (5) \end{array}$	Year (6)
Exposure	0.076^{***} (0.018)	0.129^{***} (0.023)	0.518^{***} (0.129)	$\begin{array}{c} 0.024^{***} \\ (0.007) \end{array}$	0.039^{***} (0.013)	$\begin{array}{c} 0.182^{***} \\ (0.030) \end{array}$
Observations R-squared	$130,351 \\ 0.433$	$130,351 \\ 0.443$	$130,351 \\ 0.455$	$108,939 \\ 0.471$	$108,939 \\ 0.478$	$108,939 \\ 0.484$

Table A6: Robustness: Outstanding Loans (Closed Sample) (2/2)

Notes. The table shows coefficients from estimating Equation 4. The dependent variable Δ log outstanding loans is
the change in one plus log outstanding loans one month, one quarter, and one year after the event. The independent
variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before
the event. We control for bank characteristics with variables from Table 1. We restrict the sample to borrowers
who have credit relationships with banks in the three events. Errors are clustered at bank level. ***, **, * denote
significance at the 1% , 5% , and 10% levels, respectively.

(c)	QE
(0)	ч с —

			Relationshi	ip ends $= 1$		
		Firms			Households	
	Month (1)	Quarter (2)	Year (3)	Month (4)	$\begin{array}{c} \text{Quarter} \\ (5) \end{array}$	Year (6)
Exposure	-0.001 (0.001)	-0.003 (0.002)	-0.010 (0.008)	0.001 (0.002)	-0.0002 (0.003)	-0.003 (0.006)
Observations R-squared	$241,\!311 \\ 0.405$	$241,\!311 \\ 0.409$	$241,311 \\ 0.427$	$242,823 \\ 0.4413$	$242,823 \\ 0.4442$	$242,823 \\ 0.4667$
			(b) OMT			
			Relationshi	ip ends $= 1$		
		Firms			Households	
	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)
Exposure	0.0003 (0.002)	-0.001 (0.004)	-0.005 (0.009)	-0.002 (0.002)	-0.002 (0.003)	0.0004 (0.005)
Observations R-squared	$233,402 \\ 0.407$	$233,402 \\ 0.413$	$233,402 \\ 0.43$	$231,\!840$ 0.4523	$231,\!840 \\ 0.4615$	$231,\!840$ 0.4741

Table A7: Robustness: Termination of Credit Relationships (1/2)

(a) PSI

			Relationshi	p ends = 1		
_		Firms			Households	
_	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)
Exposure	-0.006^{***} (0.001)	-0.010^{***} (0.002)	-0.039^{***} (0.008)	-0.003 (0.002)	-0.005^{**} (0.002)	-0.019^{***} (0.002)
Observations R-squared	$209,713 \\ 0.408$	$209,713 \\ 0.418$	$209,713 \\ 0.43$	$204,636 \\ 0.4569$	$204,\!636 \\ 0.4656$	$204,\!636 \\ 0.477$

Table A7: Robustness: Termination of C	Credit Relationships $(2/2)$
--	------------------------------

(c)	OE

Notes. The table shows coefficients from estimating Equation 4. The dependent variable Relationship ends = 1 is an indicator variable that is equal to 1 if the borrower does not have a credit relationship with the bank one month, one quarter, and one year after the event. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. We control for bank characteristics with variables from Table 1. Errors are clustered at bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

			()			
			Relationshi	ip ends $= 1$		
_		Firms			Households	
	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)
Exposure	-0.002 (0.001)	-0.003 (0.002)	-0.012 (0.007)	$0.003 \\ (0.002)$	0.003 (0.004)	0.001 (0.008)
Observations	314,380	314,380	314,380	438,359	438,359	438,359
R-squared	0.011	0.012	0.017	0.02	0.035	0.072
			(b) OMT			
			Relationshi	ip ends $= 1$		
		Firms			Households	
	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)
Exposure	0.0001 (0.002)	-0.001 (0.004)	-0.006 (0.010)	-0.0002 (0.003)	0.0008 (0.004)	0.004 (0.005)
Observations R-squared	$306,810 \\ 0.011$	$306,810 \\ 0.012$	$306,810 \\ 0.016$	$427,904 \\ 0.026$	$427,904 \\ 0.045$	$427,\!904$ 0.079

Table A8: Robustness: Relationship Terminations Without Borrower FE (1/2)

(a) PSI

	Relationship ends $= 1$						
-	Firms			Households			
	Month (1)	Quarter (2)	Year (3)	Month (4)	$\begin{array}{c} \text{Quarter} \\ (5) \end{array}$	Year (6)	
Exposure	-0.007^{***} (0.002)	-0.013*** (0.002)	-0.030^{***} (0.004)	-0.003 (0.003)	-0.005^{**} (0.002)	-0.017*** (0.003)	
Observations R-squared	$266,761 \\ 0.007$	$266,761 \\ 0.009$	$266,761 \\ 0.021$	$401,652 \\ 0.021$	$401,\!652 \\ 0.036$	$401,\!652 \\ 0.062$	

Table A8: Robustness: Relationship Terminations Without Borrower FE (2/2)

(c) QE

Notes. The table shows coefficients from estimating Equation 4, excluding the borrower fixed effect θ_j . The dependent variable Relationship ends = 1 is an indicator variable that is equal to 1 if the borrower does not have a credit relationship with the bank one month, one quarter, and one year after the event. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. We control for bank characteristics with variables from Table 1. For firms (Columns 1 to 3), we use 1 plus the logarithm of firm assets and firm workers, EBITDA/Assets, equity ratio, and an indicator variable that is equal to 1 if the firm has 90-day overdue loans as additional control variables. We exclude firms without balance sheet data. Errors are clustered at bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

			Accepted an	plication - 1		
_		Firms	necepted ap	pheation – 1	Households	
	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)
Exposure	0.043^{***} (0.015)	0.022^{*} (0.013)	0.023^{**} (0.009)	-0.022 (0.014)	-0.007 (0.016)	-0.002 (0.006)
Observations R-squared	$4,471 \\ 0.432$	$3,573 \\ 0.449$	$\begin{array}{c}14,\!676\\0.45\end{array}$	$3,258 \\ 0.445$	$3,116 \\ 0.457$	$13,051 \\ 0.456$
			(b) OMT			
			Accepted ap	plication = 1		
_		Firms			Households	
	Month (1)	Quarter (2)	Year (3)	Month (4)	$\begin{array}{c} \text{Quarter} \\ (5) \end{array}$	Year (6)
Exposure	$0.011 \\ (0.021)$	-0.004 (0.012)	-0.001 (0.010)	-0.024** (0.011)	-0.005 (0.012)	-0.011 (0.009)
Observations R-squared	$2,589 \\ 0.449$	$3,424 \\ 0.452$	$16,279 \\ 0.459$	$2,908 \\ 0.432$	$3,096 \\ 0.471$	$13,477 \\ 0.454$

Table A9: Robustness: Loan Applications $\left(1/2\right)$

(a) PSI

Table A9:	Robustness:	Loan A	opplications	(2)	(2)	
1010101100	100000000000000000000000000000000000000			\ _ /	_ /	

(c) QE								
Accepted application $= 1$								
	Firms			Households				
	Month (1)	Quarter (2)	Year (3)	Month (4)	$\begin{array}{c} \text{Quarter} \\ (5) \end{array}$	Year (6)		
Exposure	0.037^{***} (0.008)	0.024^{***} (0.008)	0.036^{***} (0.008)	0.013 (0.018)	-0.006 (0.015)	0.010 (0.011)		
Observations R-squared	$6,549 \\ 0.492$	$5,654 \\ 0.494$	$19,861 \\ 0.456$	$3,354 \\ 0.454$	$3,742 \\ 0.451$	$16,542 \\ 0.461$		

.pp. s(z/z)

Notes. The table shows coefficients from estimating Equation 4. The dependent variable Accepted application = 1 is an indicator variable that is equal to 1 if the loan application is accepted. Month, Quarter, and Year report coefficients for applications 0-1, 2-3, and 4-12 months after the event, respectively. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. We control for bank characteristics with variables from Table 1. Errors are clustered at bank level. ***, **, \ast denote significance at the 1%, 5%, and 10% levels, respectively.

			Accepted app	olication = 1		
-		Firms				
	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)
Exposure	0.028^{***} (0.009)	0.018^{*} (0.010)	0.017^{*} (0.009)	-0.012 (0.025)	$0.002 \\ (0.018)$	-0.009 (0.025)
Observations R-squared	$30,339 \\ 0.047$	$27,600 \\ 0.053$	$110,108 \\ 0.042$	$26,064 \\ 0.08$	$25,042 \\ 0.091$	$103,477 \\ 0.124$
			(b) OMT			
			Accepted app	olication = 1		
_		Firms			Households	
-	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)
Exposure	0.003 (0.016)	0.001 (0.009)	-0.000006 (0.011)	-0.046 (0.033)	-0.040 (0.035)	-0.025 (0.022)
Observations R-squared	$21,426 \\ 0.049$	$25,227 \\ 0.041$	$118,261 \\ 0.033$	$22,364 \\ 0.119$	$23,868 \\ 0.147$	$106,\!496 \\ 0.131$

Table A10: Robustness: Loan Applications Without Borrower FE (1/2)

(a) PSI

	Accepted application $= 1$							
_	Firms			Households				
_	Month (1)	Quarter (2)	Year (3)	Month (4)	Quarter (5)	Year (6)		
Exposure	$0.016 \\ (0.013)$	0.009 (0.012)	0.016 (0.012)	-0.022 (0.026)	-0.021 (0.022)	-0.009 (0.024)		
Observations R-squared	$44,772 \\ 0.036$	$\begin{array}{c} 37,952\\ 0.03\end{array}$	$139,384 \\ 0.041$	$23,487 \\ 0.104$	$24,885 \\ 0.075$	$109,966 \\ 0.086$		

Table A10: Robustness: Loan Applications Without Borrower FE (2/2)

(c)	QE

Notes. The table shows coefficients from estimating Equation 4, excluding the borrower fixed effect θ_j . The dependent variable Accepted application = 1 is an indicator variable that is equal to 1 if the loan application is accepted. Month, Quarter, and Year report coefficients for applications 0-1, 2-3, and 4-12 months after the event, respectively. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. We control for bank characteristics with variables from Table 1. For firms (Columns 1 to 3), we use 1 plus the logarithm of firm assets and firm workers, EBITDA/Assets, equity ratio, and an indicator variable that is equal to 1 if the firm has 90-day overdue loans as additional control variables. We exclude firms without balance sheet data. Errors are clustered at bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

Table A11: Robustness: Lending and Capital Gains (IV mod	lel)
--	------

	Δ log loans						
		Firms					
Variable	PSI (1)	OMT (2)	QE (3)	PSI (4)	$\begin{array}{c} \text{OMT} \\ (5) \end{array}$	$\begin{array}{c} ext{QE} \\ ext{(6)} \end{array}$	
Capital gains	-154.304 (8,579.068)	1.427 (2.410)	24.929^{**} (9.521)	-1.055 (1.512)	0.243 (0.700)	8.660^{***} (1.903)	
Observations	241,311	233,402	209,713	242,823	231,840	204,636	

Notes. The table shows coefficients from estimating Equation 7. The dependent variable Δ log loans is the change in log one plus outstanding loans one year after the event. The independent variable Capital gains is the change in log bank net worth induced by realized capital gains. The independent variables is defined with more detail in the text. We control for bank characteristics with variables from Table 1. Errors are clustered at the bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

	Δ log exposure pre-trend						
	Firms			Households			
_	$\begin{array}{c} \mathrm{PSI} \\ (1) \end{array}$	OMT (2)	QE (3)	PSI (4)	$\begin{array}{c} \text{OMT} \\ (5) \end{array}$	$\begin{array}{c} ext{QE} \\ ext{(6)} \end{array}$	
Exposure	0.164^{**} (0.067)	0.011 (0.059)	0.127^{*} (0.068)	0.217^{**} (0.105)	0.019 (0.042)	-0.006 (0.055)	
Observations R-squared	$241,311 \\ 0.553$	$233,402 \\ 0.540$	$209,713 \\ 0.551$	$242,823 \\ 0.655$	$231,\!840 \\ 0.659$	$204,\!636 \\ 0.624$	

Table A12: Robustness: Credit Exposure Before Events

(a) Outstanding Loans

(b) Termination of Credit Relationships

	Relationship ends $= 1$ pre-trend					
_	Firms			Households		
	$\begin{array}{c} \mathrm{PSI} \\ (1) \end{array}$	$\begin{array}{c} \text{OMT} \\ (2) \end{array}$	QE (3)	PSI (4)	$\begin{array}{c} \text{OMT} \\ (5) \end{array}$	$\begin{array}{c} ext{QE} \\ ext{(6)} \end{array}$
Exposure	0.012^{**} (0.006)	-0.002 (0.004)	0.001 (0.005)	0.029^{**} (0.012)	$0.006 \\ (0.005)$	-0.005 (0.007)
Observations R-squared	$241,311 \\ 0.569$	$233,402 \\ 0.559$	$209,713 \\ 0.568$	$242,823 \\ 0.676$	$231,\!840 \\ 0.675$	$204,\!636 \\ 0.641$

Notes. The table shows coefficients from estimating Equation 4. In Table A12a, the dependent variable is the change in lending between one year before the event and the month before the event. In Table A12a the dependent variable is an indicator variable with value equal to 1 if the borrower-bank relationship does not exist one year before the event. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. We control for bank characteristics with variables from Table 1. Errors are clustered at bank level. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

Table A13: Robustness: Loan Exposure Correcting for Pretrends

	Δ log exposure							
_	Firms			Households				
_	$\begin{array}{c} \mathrm{PSI} \\ (1) \end{array}$	OMT (2)	QE (3)	PSI (4)	$\begin{array}{c} \text{OMT} \\ (5) \end{array}$	QE (6)		
Exposure	0.097 (0.081)	$0.063 \\ (0.087)$	$\begin{array}{c} 0.330^{***} \\ (0.085) \end{array}$	$0.035 \\ (0.035)$	0.012 (0.035)	$\begin{array}{c} 0.110^{***} \\ (0.017) \end{array}$		
Observations R-squared	$241,311 \\ 0.435$	$233,402 \\ 0.436$	$209,713 \\ 0.430$	$242,823 \\ 0.461$	$231,\!840 \\ 0.468$	$204,636 \\ 0.472$		
(b) Termination of Credit Relationships								
	Relationship ends $= 1$							
_	Firms			Households				
_	$\begin{array}{c} \mathrm{PSI} \\ (1) \end{array}$	OMT (2)	QE (3)	PSI (4)	$\begin{array}{c} \text{OMT} \\ (5) \end{array}$	$\begin{array}{c} ext{QE} \\ ext{(6)} \end{array}$		
Exposure	-0.010 (0.008)	-0.005 (0.009)	-0.039^{***} (0.008)	-0.007 (0.007)	-0.0005 (0.005)	-0.018*** (0.002)		
Observations	241,311	233,402	209,713	242,823	231,840	204,636		

Notes. The table shows coefficients from estimating Equation 4. The dependent variable Δ log loan exposure is the change in log outstanding loans one year after the event. The dependent variable Relationship ends = 1 is an indicator variable that is equal to 1 if the borrower does not have a credit relationship with the bank one year after the event. The independent variable Exposure is the standardized ratio of exposed securities to total assets, measured in the last month before the event. We control for bank characteristics with variables from Table 1. In Table A13a we include the change in log outstanding loans between one year before the event and one month before the event as an additional control variable. In Table A13b, we include a dummy that is equal to 1 for credit relationships that do not exist one year before the event as an additional control variable. Errors are clustered at bank level. ***, **, ** denote significance at the 1%, 5%, and 10% levels, respectively.

0.430

0.471

0.480

0.478

0.427

0.430

R-squared

A Effect of PSI on Portuguese bond yields

In this section we measure the effect of the Greek debt restructuring (PSI) on Portuguese government bond yields. We follow a methodology that is similar to the one used by Krishnamurthy et al. (2018) to measure the impact of OMT on yields of government bonds in periphery Euro Area countries.

There was a series of events that signalled an increase in the probability of a successful debt restructuring between February and April 2012. We obtain a list of these events using Factiva. We search for all news between February 2012 and April 2012 that include the terms "Greece" and "PSI". We filter the results to include only news from Reuters. Using this procedure, in Table A14 we obtain a list of news articles that indicate an increase in the probability of a successful Greek debt restructuring.

In Table A15 we estimate the following equation:

$$\Delta yield_t = \beta \mathbb{1}_t^{PSI} + \epsilon_t \tag{11}$$

where $\Delta yield_t$ is the change in yield between day t - 1's close and day t's close, and $\mathbb{1}_t^{PSI}$ is an indicator equal to 1 on the day of the news reported in Table A14 and on the subsequent working day. We retrieve yields for the 2-year, 5-year, and 10-year reference Portuguese government bonds from MarketWatch.

On average, Portuguese bond yields fall on days with positive news about the Greek debt restructuring. We compute the cumulative change in yields on days with positive news about PSI. Yields fall by about 420 bps at the 2-year maturity, 230 bps at the 5-year maturity, and 160 bps at the 10-year maturity. These yield drops are larger than the ones reported by Krishnamurthy et al. (2018) for the OMT event (74 bps for the 2-year bond, 152 for the 5-year bond, 118 bps for the 10-year bond).

Date	Title	Summary	
4/25/2012	Greece says final participation rate in bond swap is 96.9 pct	Greek bond swap completed with 96.9% participation rate	
3/9/2012	Gilts dip as Greek deal con- firmed, eye UK data	A sufficient number of private creditors agreed on PSI conditions for a successful deal (86% participation rate)	
3/8/2012	EURO GOVT-Italian, Span- ish yields fall on Greek hopes	Yields of periphery countries went down with increased optimism of a successful Greek debt swap	
2/21/2012	ANNOUNCEMENT-Greece launches debt swap plan (PSI)	Greece announced the final terms of the debt swap program	
2/20/2012	Deal near to lower Greek debt- Greek finmin source	European countries appeared to agree on credit conditions to finance the Greek debt swap	
2/13/2012	Greek bond swap seen wrapped up in March- govt	The Greek government announced that the Greek debt swap would be completed in March 2012	
2/7/2012	EURO GOVT-Bunds slide as markets anticipate Greek deal	Yields of periphery countries went down with positive signs of a Greek debt swap deal	
2/5/2012	France says Greek PSI talks going "relatively well"	French finance minister said talks on the Greek debt swap were moving relatively well	

Table A14: List of PSI News

Notes. The table lists news between February and April 2012 that were associated with an increase in the probability of a successful Greek debt restructuring. We obtain this list using the procedure described in Appendix A

Variable	(1)	$\begin{array}{c} \Delta \text{ yield (bps)} \\ (2) \end{array}$	(3)
PSI event = 1	-34.625^{***}	-19.267**	-13.108^{**}
	(11.163)	(8.001)	(5.384)
Cumulative effect (bps)	-415.500	-231.204	-157.296
N	269	269	269
R-squared	0.026	0.009	0.014

Table A15: Regression: Effect of PSI News on Yield Changes

Notes. The table shows coefficients from estimating Equation 11 using yields of the reference 2-year, 5-year, and 10-year Portuguese government bonds. We retrieve yields measured at each day's close from MarketWatch for all days with available data in 2012. *Cumulative effect (bps)* measures the sum of daily yield changes for all days in which the indicator variable associated with the PSI event is equal to 1. We display heteroskedasticity-robust standard errors in parentheses. ***, **, * denote significance at the 1%, 5%, and 10% levels, respectively.

Acknowledgements

We thank Banco de Portugal (BPLim) for providing access to the data. We thank Viral Acharya, Juliane Begenau, Diana Bonm, Matteo Maggiori, Holger Mueller, Steven Ongena, Francisca Rebelo, Anthony Saunders, Philipp Schnabl, Amit Seru for comments and discussions. We also thank seminar participants at New York University and Stanford University.

Any errors or omissions are the responsibility of the authors. The contribution by Gil Nogueira to this paper has been prepared under the Lamfalussy Fellowship Program sponsored by the European Central Bank. Any views expressed are only those of the authors and do not necessarily represent the views of the ECB, the Banco de Portugal or the Europystem.

Laura Blattner

Stanford University, Stanford, United States; email: lblattner@stanford.edu

Luísa Farinha

Banco de Portugal, Lisbon, Portugal; email: lfarinha@bportugal.pt

Gil Nogueira

NYU Stern, New York City, United States, email: anogueir@stern.nyu.edu

© European Central Bank, 2021

Postal address 60640 Frankfurt am Main, Germany Telephone +49 69 1344 0 Website www.ecb.europa.eu

All rights reserved. Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the ECB or the authors.

This paper can be downloaded without charge from www.ecb.europa.eu, from the Social Science Research Network electronic library or from RePEc: Research Papers in Economics. Information on all of the papers published in the ECB Working Paper Series can be found on the ECB's website.

PDF ISBN 978-92-899-4860-9 ISSN 1725-2806 doi:10.2866/200470

QB-AR-21-098-EN-N